Author's response to reviews

Title: Profiling of phenolic compounds and their antioxidant and anticancer activity in (Pandanus amaryllifolius Roxb) extract from different locations of Malaysia

Authors:

   Ali Ghasemzadeh (alighasemzadeh@putra.upm.edu.my)
   Hawa ZE Jaafar (Hawazeji@agri.upm.edu.my)

Version: 5  Date: 29 October 2013

Author's response to reviews: see over
Dear Dr Tom Rowles
Executive Editor
BMC Complementary and Alternative Medicine

Thank you very much for your e-mail. The voucher specimens of samples were added in material and method section (Page 5, line 8-12).

Best regards

Ali Ghasemzadeh, Ph.D
Department of Crop Science
Faculty of Agriculture
University Putra Malaysia
43400, Serdang, Malaysia
Tel: +6 0-17-291-7679
Fax: +60-3-8943 5973